Author’s response to reviews

Title: Association of Maternal Ethnicity and Urbanicity on Severe Pediatric Disease: A Nationwide Cohort Study

Authors:

Ya-Ting Chang (nancy3681@gmail.com)
Huei-Shyong Wang (wangs444@cgmh.org.tw)
Jia-Rou Liu (jiarou@mail.cgu.edu.tw)
Chi-Nan Tseng (chinan.tseng@cgmh.org.tw)
I-Jun Chou (chouijun@gmail.com)
Shue-Fen Luo (lsf00076@adm.cgmh.org.tw)
Chang-Fu Kuo (zandis@gmail.com)
Lai-Chu See (lichu@mail.cgu.edu.tw)

Version: 2 Date: 29 Nov 2019

Author’s response to reviews:

Dear Amelia:

We are happy that our manuscript "Association of Maternal Ethnicity and Urbanicity on Severe Pediatric Disease: A Nationwide Cohort Study" (BPED-D-19-01007R1) " is accepted by your journal. The requests have been carried out:

-- Please ensure that the abstract contains only the following headings: Background, Methods, Results and Conclusions.
Ans: Yes

-- Please insert the heading 'Declarations' for the Declarations section, above the 'Abbreviations' section of your manuscript.
Ans: Yes

-- In the section 'Funding', please also describe the role of the funding body/bodies in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript.
Ans: Yes, we added "Funding body of this study did not play any role in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript".

-- We notice that author CNT is missing from the authors' contributions section. The individual contributions of all authors to the manuscript should be specified in the Authors’ Contributions section. Guidance and criteria for authorship can be found here:
http://www.biomedcentral.com/submissions/editorial-policies#authorship
Ans: Yes, we specified CNT’s contribution in conceptualizing and designing the study.

-- At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.
Ans: Yes